Author's response to reviews

Title: Domestic water carrying and its implications for health: a review and mixed methods pilot study in Limpopo Province, South Africa

Authors:

Jo-Anne L Geere (jo.geere@uea.ac.uk)
Paul R Hunter (paul.hunter@uea.ac.uk)
Paul Jagals (jagalsp@tut.ac.za)

Version: 6 Date: 25 August 2010

Author's response to reviews:

Thank-you for your email.

We have:

1. Deleted the two pictures with people on.
2. Put hyphens between the numbers and checked the final version of the manuscript
3. Removed track changes and checked the manuscript.